Author's response to reviews

Title: Understanding implementation processes of clinical pathways and clinical practice guidelines in pediatric contexts: A study protocol

Authors:

Shannon D Scott (shannon.scott@ualberta.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Terry P Klassen (tklassen@mich.ca)
Alberto Nettel-Aguirre (alberto.nettel-aguirre@albertahealthservices.ca)
David W Johnson (david.johnson@albertahealthservices.ca)

Version: 3 Date: 22 December 2011

Author's response to reviews:

Dear editing staff,

We have provided a revised manuscript with the following changes:

1. in the reference list, all of the 'et al' entries have been replaced with the actual authors names.
2. one duplicate reference entry has been removed and the reference numbers in text have been adjusted accordingly.
3. the last section of the abstract has been added.
4. the vertical lines in the table have been removed.
5. the 2 additional files have been removed.
6. all highlighting and tracking have been removed and the manuscript has been checked for typographical errors.

Thank-you for the opportunity to publish with your well respected publication!

The authors.